Author's response to reviews

Title: Development and Validation of Psychosocial Determinants Measures of Physical Activity among Iranian Adolescent Girls

Authors:

Ashraf Pirasteh (pirasteh2002@yahoo.com)
Alireza Hidarnia (hidarnia@modares.ac.ir)
Ali Asghari (asghari_a@yahoo.com)
Soghrate Faghihzadeh (faghihz@modares.ac.ir)
Fazlollah Ghofranipour (ghofranf@modares.ac.ir)

Version: 2 Date: 9 September 2007

Author's response to reviews:

Development and Validation of Psychosocial Determinants Measures of Physical Activity among Iranian Adolescent Girls

Pirasteh A, Hidarnia AR*, Asghari A, Faghihzadeh S, Ghofranipor F,

* Corresponding author. Department of Health Education, Tarbiat Modares University, Tehran 14115 111, IRAN, Tel: +98-21-88011001; Fax: +98-21-88013030
E-mail: hidarnia@modares.ac.ir

Ashraf Pirasteh
Ph.D Candidate in Health Education
Tarbiat Modares University
Tehran, 14115-111, Iran
Tel: +98-21-88011001; Fax: +98-21- 88013030
E-mail: pirasteh@modares.ac.ir

Alireza Hidarnia, Associate Professor of Health Education
Department of Health Education
Tarbiat Modares University

Soghrate Faghihzadeh, Associate Professor of Vital Statistics
Department of Vital Statistics
Dear Editor

Greetings.

We would like to submit our paper entitled: "Development and Validation Psychosocial Determinants Measures of Physical Activity among Iranian Adolescent Girls" for publication in the BMC Public Health. This study is a Ph.D dissertation. I would appreciate it if you consider this manuscript for evaluation at your earliest convenience. We look forward to hearing from you.

The address of the author responsible for communication regarding the manuscript is:

Alireza Hidarnia

Academic Degree: Associate Professor

Address: Department of Health Education Tarbiat Modares University, Tehran, Iran

Tel: +9821 88011001 (ext: 3550, 3553)

Fax: +9821 88006544

E-mail: hidarnia@modares.ac.ir

With Kindest Regards

Sincerely Yours,

Alireza Hidarnia